

RESEARCH ARTICLE

OPEN ACCESS

Evaluation of *Fusobacterium nucleatum* in Amniotic Fluid of Women with Preterm Labour

Fusobacterium nucleatum in Amniotic Fluid of Women with Preterm Labour Nwaokorie FO^{1ID}, Okunade KS^{2ID}, Edet UO^{3ID}, Obisanya OI^{1ID}, Ekun O^{1ID}

¹Department of Medical Laboratory Science, College of Medicine of the University of Lagos, Idi-araba, Lagos, Nigeria

²Department of Obstetrics and Gynaecology, College of Medicine of the University of Lagos/Lagos University Teaching Hospital, Idi-araba, Lagos, Nigeria

³Department of Biological Science, Faculty of Natural and Applied Sciences, Arthur Jarvis University, Cross River State, Akpabuyo, Nigeria

Submitted: 7th August 2023 Accepted: 21st September 2023 Published: 31st December 2023 ID: Orcid ID

Abstract

Objective: The effect of *Fusobacterium nucleatum*, an oral commensal, on pregnancy outcomes is of global concern. This study screened for *F. nucleatum* and the *fadA* gene in amniotic fluid obtained from women presenting with preterm labour.

Methods: This is a cross-sectional pilot study. Amniotic fluid was collected from women with preterm labour at Lagos University Teaching Hospital and analysed using conventional polymerase chain reaction (PCR). I6SrRNA species-specific primers were used to assess the presence of *F. nucleatum* species and *fadA* genes for virulence potentials. Questionnaire-based data obtained was analysed using SPSS version 20.0. and amplicon analysis was measured on base pairs.

Results: Fourty participants had preterm deliveries between April and August 2018. Of these 34 (85%) were enrolled, out of which 25 amniotic fluid samples (AF) were collected. The mean age of the participants was 30.5 years. Among them, 10 (29.4%) were primigravida, and 24 (70.6%) were multigravida. Six (1.8%) of the women had a history of gestational diabetes, and four (1.2%) had hypertension. Only 4 (1.2%) were aware that bacteria could cause preterm labour. 11 (32%) visited the dentist during the period of their pregnancy. There was no amplification of 16SrRNA or the fadA genes in the tested samples.

Conclusions: This study showed that preterm delivery was more common in multigravida pregnant women aged 31-35 years. Although some of the samples were cloudy and purulent, *Fusobacterium nucleatum* species-specific and virulent *fadA* genes were not detected in amniotic fluid obtained from the population of women with preterm labour studied.

Keywords: Anaerobes, *Fusobacterium nucleatum*, Preterm labour, Primigravida, Multigravida, Amniotic fluid, Polymerase chain reaction

Plain English Summary

Microbiological analysis reveals the presence of infectious agents in preterm deliveries. This study used a molecular-based method to analyse the presence of an anaerobe *F. nucleatum*, known for causing preterm delivery in women, and its virulence gene potential. The study did not detect the species of interest but created awareness about the involvement of microbes in preterm delivery. It also highlighted that preterm delivery was seen more in multigravida pregnant women aged 31-35 years.

Background

Preterm delivery is defined as birth before 37 weeks of gestation (1). Studies have shown that

Correspondence: Nwaokorie, Francisca O Department of Medical Laboratory Science, College of Medicine of the University of Lagos, Idi-araba, Lagos, Nigeria. +2348037173766, fnwaokorie@unilag.edu.ng risk factors range from immunological to genetic disorders, endocrine disorders and preeclampsia (1, 2, 3). Others are lifestyle and

© BUMJ. 2023 Open Access This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/bv/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/ublicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated.

behavioural factors, uterine leiomyoma, family history, the interval between pregnancy, and medical, gynaecological and obstetric risk factors (3). According to WHO's key fact (4), an estimated 15 million babies are born preterm annually. Neonatal mortality contributes to under-five mortality, including the contribution of preterm neonatal mortality. The burden exerted by Nigeria as a nation with the second-highest number of neonatal deaths worldwide cannot be overlooked.

Globally, the prevalence of preterm birth ranges from $9\cdot0-12\cdot0\%$, with Europe, India, China, Nigeria, Bangladesh, and Indonesia accounting for a greater percentage of the burden in that order (5). The prevalence rate associated with preterm delivery in Enugu, Nigeria, is 16.9% (6), 11.8% in llorin (7) and 8% in Lagos (8). None of these studies reported microbiological analysis or laboratory-specific diagnostic findings on the contributions of infectious agents to preterm delivery.

Maternal infections (16.6%) account for the second highest risk factor of preterm delivery next to maternal malnutrition (17.5%) in low- and middle-income countries (9). Reported infections are mainly those of HIV, malaria, syphilis, chlamydia, asymptomatic bacteriuria, periodontal Infection, bacterial vaginosis and Group B streptococcal colonization. Periodontal diseases are worse during pregnancy due to the presence of hormonal imbalances. A good percentage of women report excessive gum bleeding during pregnancy due to pregnancyassociated gingivitis, and plaque-induced inflammation of the gingiva (10). In Ibadan, Opeodu et al. (11) demonstrated women's periodontal condition and treatment needs during pregnancy. The fact that 89.3% of the women had never visited a dentist in their lives and observed conditions showed the need for preventive dental visits. The periodontal pathogens responsible for this undergo haematological translocation from the oral cavity extra-oral infections to cause (12). Fusobacterium nucleatum is an oral commensal associated with gingivitis, periodontitis, and oral cancer (13, 14). It is implicated in metastatic infections involving the heart valves (15), liver brain, ioints (17). and septic (16).thrombophlebitis in extra-pharyngeal locations (2), and the uterus (14), It is also isolated commonly in intrauterine infections (14).

Periodontitis is caused by bacteria that produce significant amounts of proinflammatory mediators, mainly IL-1, IL-6, PGE2, and TNF- α (18). *F. nucleatum* has the characteristic ability to locate, adhere and rapidly colonize (19, 20) to elude the host's defence mechanisms and to produce substances that directly contribute to

the destruction of tissues (21). They can translocate to placenta sites, or the by-products they produce may reach the placenta and enter the amniotic fluid and fetal circulation, serving to activate inflammatory signalling pathways. This may promote labour activation through the production of placental and chorion–amnion prostaglandin E2 (PGE2) (22). Women with preterm birth (PTB) demonstrated significantly increased gingival crevicular fluid (GCF) levels of IL-6 and PGE2 compared with those who had full-term births (23).

Specifically. F. nucleatum stimulates arachidonic acid metabolism, a rate-limiting step for prostaglandin synthesis, in the human uterine endometrium. These metabolites are important mediators of inflammation. In addition, F. nucleatum harbours the conserved Fusobacteium Adhesin A (fadA) that encodes the host colonisation factor. Additionally, it produces phospholipases that play a role in remodelling the stability of lipid membranes, therefore affecting placental and foetal development in pregnancy (24). The FadA and N-Utilisation Substances (nusG) virulence genes peculiar to pathogenic strains of Fusobacterium species have been studied and linked as having major virulent potential (15). Recent investigations have demonstrated that Fusobacterium nucleate colonizes the placenta by binding its Fap2 lectin to the placentally displayed Gal-GalNAc (25).

Several risk factors have been associated with preterm and low-weight births, and these include genitourinary and periodontal infections (26). Another important risk factor is the distinct bacteria communities that are associated with the placental tissues of patients with severe chorioamnionitis. *Fusobacterium* species have been detected in the placenta tissues of women who had severe chorioamnionitis (27, 28, 29). These findings suggest a possible association between bacteria like *Fusobacterium nucleatum* and complications such as miscarriage, intrauterine death, neonatal death, neonatal sepsis, preterm delivery, and premature rupture of membranes (27, 28, 29).

Since the adverse effect of microorganisms on pregnancy outcomes was revealed, this opinion is being supported by several reviews and research (2, 10, 14, 30, 31, 32). However, such information is limited in Nigeria. Looking at the prevalence of oral infection estimated to be 15% - 58% in females of reproductive age and the concern over poor oral health hygiene (11, 33) as well as a high rate of preterm delivery (6, 7, 8), it would not be out of place to investigate the involvement of microbial pathogens in maternal infection that could lead to preterm birth. This research and diagnostic investigation aspect may have been neglected during antenatal care. Creating awareness and demonstrating the involvement of microbial agents will provide more insight into the need to consider microbial investigations as a way to highlight microbial infection of the placenta as a risk factor for preterm birth and therefore guide strategies to mitigate its associated adverse pregnancy outcomes.

Materials and Methods

This is a pilot cross-sectional study of consenting pregnant women with preterm labour and delivery between 28 and 36 weeks gestation from April to August 2018.

Study site

Lagos University Teaching Hospital is located in Mushin, Lagos State, Nigeria. It is one of the largest hospitals in the country and provides maternal care mostly to people in the middle and higher socio-economic classes, with referrals that cut across other socio-economic classes. The Department of Obstetrics and Gynaecology was established in 1967. They provide antenatal care and delivery management for booked patients. It also accepts unbooked patients and treats obstetric and gynaecological emergencies admitted through the Accident and Emergency Unit.

Study procedure and data collection

Pregnant women with preterm labour who gave informed consent for participation were enrolled. Women who had induced preterm delivery, antepartum haemorrhage or those delivered before 28 weeks and after 36 weeks of gestation were excluded. The women's demographic information was obtained and knowledge of oral hygiene was assessed with a staff-assisted structured questionnaire that was administered individually to the participants. This was sectioned into demographics, dental care and history, pregnancy information and assessment of knowledge of preterm labour.

Sample collection

Amniotic fluid samples were collected from the posterior fornix after inserting a sterile Cusco's speculum into the vaginal cavity to visualise the cervix and detect amniotic fluid or during caesarean section using a sterile Pasteur pipette as described by Adeyeba *et al.* (34) and then transported to the laboratory in a sterile universal bottle. There was minimal risk of discomfort, cross-contamination and infection due to the process of collecting amniotic fluid and the strict observance of asepsis throughout this process. Samples that failed to meet the criteria for DNA extraction as a result of heavy contamination from blood and meconium stains during sampling were not processed further.

Detection of F. nucleatum species-specific and fad genes by PCR DNA Extraction

To detect the presence of F. nucleatum speciesspecific and fad genes in the amniotic fluid, DNA extraction and purification were done using the Norgen Genomic DNA Isolation Kit (Model 24700, Norgen Biotek, Corp, Canada). Briefly, 150 µL of Amniotic fluid was transferred to a 1.5 mL microcentrifuge tube followed by the addition of 150 µL of Digestion Buffer A. Then, 12 µL of Proteinase K was added to the suspension. This was mixed by gentle vortexing and incubated at 55 °C for 1 hour on a dry bath Incubator (Model 190, Thermo Scientific) then 300 µL of Buffer SK and 300 μL of 96 - 100% ethanol were added to the Lysate and the mixture was vortexed. Moreso, 600 µL of the mixture was thereafter transferred into a spin column in a 2 ml collection tube and centrifuged at 8000 rpm for 3 minutes. The flow through in the collection tube was discarded and reassembled and the spinning was repeated for the remaining Lysate. Furthermore, 500 µL of Wash Solution A was added and the mixture was centrifuged at 14000 rpm for 1 min. The flow-through was discarded and the column was reassembled. Then 500 µL of Wash Solution A was added again and centrifuged at 14000 rpm for 2min. The extracted DNA was purified by placing the Norgen spin collection column into a sterile 1.7 mL Elution tube and the collection tube containing the filtrate was discarded. 200 µL of Elution Buffer B was added to the centre of the resin bed and then centrifuged at 6000 rpm for 1min and at 14000 rpm for an additional 2min to collect the total elution volume (DNA). The quality of the extracted DNA was analysed using 1.5% agarose gel electrophoresis in Tris-acetate EDTA (TAE) buffer (Containing 40 mM Trisacetate, 2 mM EDTA (Ph 8.3) performed at 70 V for 2.5 hours using 100 bp DNA ladder (Solis BioDyne 07-11-00050) as standard molecular weight marker. Before then, the DNA was stained with 6X loading dye (Thermo Scientific) for visual tracking of DNA migration during electrophoresis before loading the mixture into the gel tank. Gels were stained with 0.5 µg/ml of ethidium bromide for 45 minutes and were stained with water for 20 minutes. Stained gels were examined and photographed using a Digital Photo documentation System (Clinix, 1000).

DNA amplification

This was performed using *F. nucleatum* primers of 16S rRNA-F (5' -AGA GTT TGA TCC TGG CTC AG -3') and 16S rRNA-R (5'-GTC ATC

GTG CAC ACA GAA TTG CTG-3') to amplify a 360-bp region of the 16S rRNA gene (35) The presence of virulence fadA gene was detected using the primer pair fadA-F (5' -CAC AAG CTG ACG CTG CTA GA -3') and fadA -R (5' -TTA CCA GCT CTT AAA GCT TG -3' 9) to amplify a 232-bp region of the fadA gene (15) from positive samples of F. nucleatum. Amplifications were carried out using final volumes of 20 µL, with the reaction mix containing 12 µL sterile Milli-Q water, 1 X PCR buffer, 7.5 mM of dNTPs, Tag DNA polymerase, 0.16 µmol/L of each primer, and 3 µg of DNA. Amplification was carried out in thermocycler 005 (A& E Lab, UK) programmed to run the following cycling conditions; initial denaturing at 95 °C for 3 min; 30 cycles of denaturing at 95 °C for 30secs, annealing at 53 °C for 1 min, extension at 72 °C for 45 secs, followed by a final extension at 72 °C for 5 min. As a positive control, a reference strain F. nucleatum ATCC 10953 and Sterile Water was included as a negative control. Gel Electrophoresis and visualization

In all PCR assays, 20 μ l of the amplified products were separated using 1.5% agarose gel electrophoresis in Tri-acetate EDTA (TAE) buffer (40 mM Tris-acetate, 2 mM EDTA (Ph 8.3) performed at 70 V for 2.5 hours using 100 bp DNA ladder (Solis BioDyne) as standard molecular weight marker. Gels were stained with 0.5 μ g/ml of ethidium bromide for 45 minutes and were stained with water for 20 minutes. Stained gels were examined and photographed using a Digital Photo Documentation System (Clinix Model 1000).

Statistical analysis

The questionnaire was analysed using the Statistical Package for Social Sciences (SPSS) version 20.0. Descriptive statistics using frequencies, cross-tabulations and calculation of mean values for the variables were determined. Amplicon analysis was done by gel visualisation to observe and measure the presence or absence of amplification bands in base pairs.

Results

Out of a total of 40 participants who had preterm deliveries in LUTH from April to August 2018, 34/40 (85%) were enrolled and 25 amniotic fluid samples (AF) were collected (Figure 1). Nine of the 25 samples collected did not meet the criteria for DNA extraction. Thus, 16 samples went through PCR amplification processes and were analysed. Two of the amniotic fluids tested were cloudy and one was offensively purulent.



Figure 1: Gel analysis of amplicon from single plex –PCR. Gel Image showing electrophoresis bands. Lane 1: DNA Standard Marker, Lane (+) Positive control, Lane (-) negative control, Lanes 1-14. sample lanes show no amplification band that corresponds to the expected band sizes of **360 bp** for *F. nucleatum* 16S rRNA.

Table 1 shows the mode of delivery among women with preterm labour. Of the 16 women whose amniotic fluids were collected, 2 were by vaginal collection and 14 were during caesarean sections. The majority of the participants were between the ages of 31 and 35 years (Table 2), with a mean age of 30.5 years. Ten (29.4%) of the participants were primigravida, and 24 (70.6%) were multigravida.

Table 1: Mode of Delivery among Women with Preterm Labour

Mode of delivery	Eligible (%)	Non-eligible (%)
Vagina	2 (5.88)	9 (26.47)
Caesarean section	14 (41.18)	9 (26.47)

VARIABLE	FREQUENCY (n = 34)	Percentage
Age range		
20 – 25	7	20.6
26 - 30	9	26.5
31 – 35	18	52.9
Gravidity		
Primigravida	10	29.4
Multigravida	24	70.6

Table 2: Age and Gravidity level of Women presenting with Preterm Labour VARIABLE ERECUENCY (n = 34)

The overall assessment of the health history and knowledge of these women on causes of preterm labour shows that 6 (17.6%) of the

women had a history of diabetes, 4 (11.8%) had hypertension, and 5 (14.7%) had other pregnancy complications (Table 3).

Table 3:	Health Histor	y and Knowledge	of Women on th	e Causes of Preterm	۱ Labour (n=34)
----------	----------------------	-----------------	----------------	---------------------	-----------------

Medical history and oral health risk factors		Responses (%)	
	Yes	No	
The number who had a history of diabetes	6 (17.6)	28 (82.4)	
The number who had a history of Hypertension	4 (11.8)	32 (88.2)	
The number who had pregnancy complication(s)	5 (14.7)	29 (85.3)	
Do you experience any toothache or pain around your gum?	15 (44.1)	19 (55.9)	
Did you visit a dentist during this pregnancy?	11(32.4)	23 (67.4)	
Do you know that germs (bacteria) in your mouth can cause preterm labour?	4 (11.8)	32 (88.2)	

Only 4 (11.8%) were aware that bacteria could cause preterm labour, and 11 (32.4%) of the participants had visited the dentist during pregnancy. Of the 16 samples amplified none

had amplification corresponding to expected band sizes of 360 bp for *F. nucleatum* 16SrRNA (Plate 1) and 232 bp for the *fadA* gene (Table 4).

Table 4: PCR analysis of Amniotic Fluid from women with Preterm Labour at LU
--

S/n	DNA Code	Macroscopy	PCR result	
			FN16S gene	FadA gene
1	K001	Meconium stained	No amplification	No amplification
2	K002	Purulent, Cloudy yellow	63	63
3	K003	Cloudy	63	0
4	K004	Cloudy	63	63
5	K005	Clear pale yellow	63	63
6	K006	Clear pale yellow	63	0
7	K007	Clear pale yellow	63	63
8	K008	Clear	63	0
9	K009	Clear	63	63
10	K010	Clear	63	63
11	K011	Clear	63	63
12	K012	Clear pale yellow	63	63
13	K013	Clear pale yellow	63	63
14	K014	Clear	63	63
15	K015	Clear pale yellow	63	63
16	K016	Clear pale yellow	63	63
17	Positive Control (F.		Amplification	-
	nucleatum ATCC		·	
	10953)			
18	Negative Control (PCR		No amplification	No amplification
	mix and Sterile Water)		·	
	,			

Table 4 shows that none of the 16 samples was amplified except for the control.

Discussion:

Preterm birth is a worrisome complication of pregnancy that incurs short to long-term medical

and financial burdens on affected children, their families, and the health care system. For the first time in Lagos University Teaching Hospital, a

study was conducted to assess the contribution of a microbial pathogen, F. nucleatum, to preterm delivery. In modern-day laboratory medicine, genetic biomarkers are used on direct clinical specimens to detect the presence of microbial species and their virulent characteristics. The set of oligonucleotide primers for the two targeted genes used showed no amplification, indicating that there was no presence of *F. nucleatum* species in the uterus. However, our finding is at variance with the previous studies by You et al. (36) in Korea and that of Vander Haar et al. (28) in which F. nucleatum was detected in amniotic fluid samples.

None of the 16 amniotic fluids screened had *F.* nucleatum even though some of these samples were purulent and cloudy as opposed to the normal pale. clear yellow appearance of the amniotic fluid which is meant to be sterile. The primer used has been demonstrated to be species-specific for *F. nucleatum*, however, this is not enough to conclude the absence of these species as there are five (5) different subspecies of *F. nucleatum* (five: *F. nucleatum, animalis,* polymorphum, fusiforme, vincentii) (37). This has opened up the need to conduct further studies using a wide range of species-specific primers.

To ensure rapid and specific identification of F. nucleatum and its novel virulent FadA adhesin gene, these invasive samples were processed using a conventional polymerase chain reaction (PCR). This technique detects and characterizes the presence of DNA from both viable and non-viable organisms. Thus, the popular cultural method that supports the isolation and characterization of only the viable bacteria in the clinical sample was not utilized in this study. Moreover, our target species is an anaerobe and anaerobic cultures are known to be quite challenging. In addition, preterm deliveries have been linked to the virulent actions of other microbial species such as S. sanguinegens, and, Gardnerella sp. which have been detected in addition to F. nucleatum in amniotic fluid using more advanced 16S rRNA gene sequencing (36). Thus, species such as Ureaplasma urealyticum, Mycoplasma hominis, Gardnerella vaginalis, and Bacteroides (36) need to be examined in future studies exploring both cultural and genetic-based laboratory procedures.

The *FadA* virulence gene is specific to pathogenic strains of *Fusobacterium* species. This gene has been studied and linked as a major virulent potential of these species (5). The species have been detected in the sampling of dental plaque, high vaginal swabs, amniotic fluid, and chorioamnionic tissue. The presence

of *F. nucleatum* in the amniotic fluid from women with caesarean section at full term or placenta tissues has been demonstrated using Polymerase Chain Reaction (PCR) and sequencing-based technique (26, 27, 28). Other genes are coding for virulence other than Fusobacterium adhesion A (fadA). For example, the adhesin molecules (virulence) have been identified as Fatty Acid Binding Protein 2 (Fap2) (2, 25). Primer specific for this gene can also be used to evaluate the presence of an F. nucleatum virulence gene. We hope to utilize these primers in our future study. In addition, the small sample size collected and analysed might have contributed to the result. It is therefore important for future studies to examine a larger population. especially of women from other hospital environments in Nigeria as species of interest may be observed.

Age is an important risk factor for many adverse events in pregnancy (38). The study observed that most of the participants were in the age range of 31 - 35 years. This seems to be in line with the study of Mokuolu et al. in 2010 (7) who from their study on prevalence and determinants of pre-term deliveries in Nigeria showed that this pregnancy complication is common among women aged 20 - 34 years. Most of the pregnant women presenting with preterm labour in this study were multigravida. This is also unexpected as pregnancy complication such as preterm delivery has not been strongly linked with many previous pregnancies and deliveries (9, 39, 40). This was further corroborated by our findings; however, the absence of a link may have been due to the small sample size in our study.

Study limitations

There were a few limitations in this study. The study assayed a small population and thus may not be used to make any statistical inference. There is a need to conduct a future study that involves a larger population of women with preterm labour to comprehensively establish if there is the involvement of microbial pathogens in preterm delivery. The sites of amniotic fluid sample collection (vaginal versus abdominal) could potentially create some differences in the yield and characteristics of the results obtained in our study. However, very strict aseptic techniques were ensured during the procedures regardless of the site of sample collection. The study did not include the evaluation of the oral health status and microbial evaluation to study the occurrence of the target species in the oral cavity. There are about five subspecies of Fusobacterium nucleatum. The study used one primer for F. nucleatum species and virulence primers. It would be good to conduct future research using more species, subspecies and virulence-specific primers.

Conclusion

F. nucleatum species-specific and virulent *fadA* genes were not detected in amniotic fluid samples of pregnant women presenting with preterm delivery in LUTH. The absence of the oral commensal suggests the need for future studies to utilize more samples as well as multiplex primers capable of detecting more species of *Fusobacterium* and other commonly implicated pathogens in preterm delivery in Nigerian women living in the urban city of Lagos.

List of Abbreviations

DNA: Deoxyribonucleic acid

- EDTA: Ethylenediaminetetraacetic Acid
- fadA: Fusobacterium adhesion A
- HIV: Human Immunodeficiency Virus
- LUTH: Lagos University Teaching Hospital
- PCR: Polymerase Chain Reaction
- PGE2: Prostaglandin E2
- PTB: Preterm Birth
- SPSS: Statistical Package for Social Sciences
- WHO: World Health Organization

Declarations

Ethics approval and consent to participate

Ethics approval and consent to participate of Responsible conduct research was employed- during the study by obtaining Ethical approval with Reference no. ADM/DCST/HREC/APP/2289 from the Health Research Ethics Committee, Lagos University Teaching Hospital, before embarking on the research. To obtain consent and ensure ethical research the study was explained to the participants in the language they understood. They were assured that participation was free and that they could withdraw from the study whenever they wanted. Neither would their withdrawal interfere with their treatment.

Consent for publication

All the authors gave consent for the publication of the work under the Creative Commons Attribution-Non-Commercial 4.0 license. We otherwise convey all copyright ownership, including all rights incidental thereto, exclusively to the journal when published.

Availability of data and materials

Data generated in this study are contained in this manuscript.

Conflict of interests

The authors have declared that they have no competing interests.

Funding

The research was not funded.

Authors' contributions

This work was carried out in collaboration with all authors. Authors FON, KO, OIS, and OE designed the study, collected the data and wrote a draft of the manuscript. Authors KSO, FON and OIS, recruited and examined the patients. Authors FON and OIS performed the laboratory experiments and KSO performed the clinical studies. FON and-OU provided technical advice and revised the manuscript. All authors read and approved the final manuscript.

Acknowledgement

We acknowledge the support given by staff in Obstetrics and Gynaecology, the Resident Doctors on call in the labour ward, nurses and other health officers.

References

- 1. World Health Organisation. Recommendations on Interventions to Improve Preterm Birth Outcomes. World Health Organization 2015.
- 2. Han YW. *Fusobacterium nucleatum:* a commensal-turned pathogen: Curr Opin Microbiol. 2015: 10: 141–147. https://doi.org/10.1016/j.mib.2014.11.013
- 3. Field CA, Gidley MD, Preshaw PM, Jakubovics N. Investigation and quantification of key periodontal pathogens in patients with type 2 diabetes. J Periodontal Res 2012: 47:470–478. <u>https://doi.org/10.1111/j.1600-</u> 0765.2011.01455.x
- Blencowe H, Cousens S, Oestergaard M, Chou D, Moller AB., et al. National, regional and worldwide estimates of preterm birth rates in the year 2010 with time trends for selected countries since 1990: a systematic analysis. 2012: For CHERG/WHO. <u>https://doi.org/10.1016/S0140-</u> 6736(12)60820-4
- Chawanpaiboon S, Vogel JP, Moller AB, Lumbiganon P, Petzold M, and Hogan D. Global, regional, and national estimates of levels of preterm birth in 2014: a systematic review and modelling analysis. Lancet Glob Health 2019;7:e37-46. <u>https://doi.org/10.1016/S2214-</u> 109X(18)30451-0
- Iyoke CA, Lawani OL, Ezugwu EC, Ilechukwu G, Nkwo PO, Mba SG, Asinobi IN. Prevalence and perinatal mortality associated with preterm births in a tertiary medical centre in South East Nigeria. Inter J Wom Health 2014:6:881–888. <u>https://doi.org/10.2147/IJWH.S72229</u>

- Mokuolu, O.A., Suleiman, B.M., Adesiyun, O.O., Adeniyi, A. Prevalence and determinants of pre-term deliveries in the University of Ilorin Teaching Hospital, Ilorin, Nigeria. Pediatr Rep. 2010; 2:e3. <u>https://doi.org/10.4081/pr.2010.e3</u>
- Maymunah AO, Kehinde O, Abidoye G, Oluwatosin A. Hypercholesterolaemia in pregnancy as a predictor of adverse pregnancy outcome. Afr. Health Sci. 2014: 14 (4):967-973. https://doi.org/10.4314/ahs.v14i4.28
- Bryce E, Gurung S, Tong H, Katz J, Lee AC, Black RE, Walker N. Population attributable fractions for risk factors for spontaneous preterm births in 81 low- and middle-income countries: A systematic analysis. J Glob Health. 2022 Mar 26;12:04013. https://doi.org/10.7189/jogh.12.04013
- 10. Davenport ES. Preterm low birth weight and the role of oral bacteria. J Oral Microbiol. 2010; 2: 5779. https://doi.org/10.3402/jom.v2i0.5779
- 11. Opeodu OI, Dosumu EB, Arowojolu MO. Periodontal Condition and Treatment Needs of Some Pregnant Women in Ibadan, Nigeria. Ann Med Health Sci Res. 2015: 5(3); 213–217. https://doi.org/10.4103/2141-9248.157514
- 12. Han YW, Fardini Y, Chen C, Iacampo KG, Peraino VA, Shamonki JM, Redline RW. Term stillbirth caused by oral *Fusobacterium nucleatum*. Obstet Gynecol 2011: 115:442–445. <u>https://doi.org/10.1097/AOG.0b013e3181c</u> <u>b9955</u>
- Nwaokorie FO, Savage KO, Muinah A, Fowora MA, Ayanbadejo PO, Ogunsola FT, Coker AO. Detection of FUS-1 (OXA-85), a Class D Beta-lactamase from Fusobacterium nucleatum Subspecies Polymorphum in Nigeria. *Micro. Res. J. Inter*, 2013; 3(4),92–500. <u>https://doi.org/10.9734/BMRJ/2013/3599</u>
- 14. Ren H. and Du M. Role of Maternal Periodontitis in Preterm Birth. *Immunol* 2017; 8:139. https://doi.org/10.3389/fimmu.2017.00139
- Liu P, Liu Y, Wang J, Guo Y, Zhang Y, Shuiqing Xiao. Detection of *Fusobacterium nucleatum* and *fadA* Adhesin Gene in Patients with Orthodontic Gingivitis and Non-Orthodontic Periodontal Inflammation. PLoS ONE 2014; 9(1): e85280. <u>https://doi.org/10.1371/journal.pone.00852</u> 80
- Ohyama H, Nakasho K, Yamanegi K, Noiri Y, Kuhara A, Kato-Kogoe N, Yamada N, Hata M, Nishimura F, Ebisu S, Terada N. An unusual autopsy case of pyogenic liver

abscess caused by periodontal bacteria 2009;62:381–383. https://doi.org/10.7883/yoken.JJID.2009.38

- Cahill DP, Barker FG, Davis KR, Kalva SP, Sahai I, Frosch MP. Case records of the Massachusetts General Hospital. Case 10. A 37-year-old woman with weakness and a mass in the brain. N Engl J Med 2010;362:1326–1333. <u>https://doi:</u> 10.1056/NEJM192311291892226
- Phlstrom BL, Michalowicz BS, Johnson NW. Periodontal diseases. Lancet 2005; 366(9499):1809–20. <u>https://doi.org/10.1016/S0140-6736(05)67728-8</u>
- Goldenberg RL, Culhane JF, Iams JD, Romero R. Epidemiology and causes of preterm birth. Lancet 2008;371(9606):75– 84. <u>https://doi.org/10.1016/S0140-6736(08)60074-4</u>
- 20. Zi MY, Longo PL, Bueno-Silva B, Mayer MP. Mechanisms involved in the association between periodontitis and complications in pregnancy. Front Public Health 2014;2:290. https://doi.org/10.3389/fpubh.2014.00290
- Beck S, Wojdyla D, Say L, Betran AP, Merialdi M, Requejo JH. The worldwide incidence of preterm birth: a systematic review of maternal mortality and morbidity. Bull World Health Organ 2010; 88(1):31–38. https://doi.org/10.2471/BLT.08.062554
- 22. Amabebe E, Reynolds S, He X, Wood R, Stern V, Anumba DOC. Infection/ inflammation-associated preterm delivery within 14 days of presentation with symptoms of preterm labour: A multivariate predictive model. PLoS ONE 2019; 14(9): e0222455.

https://doi.org/10.1371/journal.pone.02224 55

- Perunovic N, Rakic MM, Nikolic LI, Jankovic SM, Aleksic ZM, Plecas DV. The association between periodontal inflammation and labour triggers (elevated cytokine levels) in preterm birth: a cross-sectional study. J Periodontol 2016; 87(3):248–56. https://doi.org/10.1902/jop.2015.150364
- 24. Besenboeck C, Cvitic S, Lang U, Desoye G, Wadwack C. Reproduction 2016; 151(6): R91-R102.

https://doi.org/10.1902/jop.2015.150364

25. Parhi L, Abed J, Shhadeh A, Alon-Maimon T, Udi S, Ben-Arye SL, Tam J, Parnas O, Padler-Karavani V, Goldman-Wohl D, Yagel S, Mandelboim O, Bachrach G. Placental colonization by Fusobacterium nucleatum is mediated by binding of the Fap2 lectin to placentally displayed Gal-GalNAc. Cell Rep 2022;22;38(12):110537.

https://doi.org/10.1016/j.celrep.2022.11053

- 26. Tellapragada C, Eshwara VK, Bhat P, Acharya S, Kamath A, Bhat S. Risk factors for preterm birth and low birth weight among pregnant Indian women: a hospital-based prospective study. J Prev Med Public Health 2016; 49(3):165–75. https://doi.org/10.3961/jpmph.16.022
- 27. Doyle RM, Harris K, Kamiza S, Harjunmaa U, Ashorn U, Nkhoma M. Bacterial communities found in placental tissues are associated with severe chorioamnionitis and adverse birth outcomes. *PLoS ONE* 2017;12(7): e0180167. <u>https://doi.org/10.1371/journal.pone.01801</u> <u>67</u>
- 28. Vander Haar EL, So J, Gyamfi-Bannerman C, Han YW. *Fusobacterium nucleatum* and adverse pregnancy outcomes: Epidemiological and mechanistic evidence. J Anae 2018; https://doi.org/10.1016/j.anaerobe.2018.01. 008
- 29. Wang X, Buhimschi CS, Temoin S, Bhandari V, Han YW, Buhimschi IA. Comparative microbial analysis of paired amniotic fluid and cord blood from pregnancies complicated by preterm birth and early-onset neonatal sepsis. PLoS One. 2013; 8:e56131. <u>https://doi.org/10.1371/journal.pone.00561</u> 31
- 30. Klebanoff M. and Searle K. The role of inflammation in preterm birth—focus on periodontitis. Br J Gynae 2006;113(3):43–45. <u>https://doi.org/10.1111/j.1471-0528.2006.01121.x</u>
- 31. Srinivasan U, Misra D, Marazita ML, Foxman B. Vaginal and Oral Microbes, Host Genotype and Preterm Birth. *Med Hypotheses* 2009;73(6): 963–975. <u>https://doi.org/10.1016/j.mehy.2009.06.017</u>
- 32. Ghandali NS, Ali HE, Masumeh D. The Impact of Microorganisms in Pregnancy. Asian Journal of Medicine and Health 2021; 22-38. <u>https://doi.org/10.9734/ajmah/2021/v19i530</u> 326
- 33. Akpata ES. Oral health in Nigeria. Intern Dent J. 2004; 54:361-366. <u>https://doi.org/10.1111/j.1875-</u> 595X.2004.tb00012.x
- 34. Adeyeba OA, Adeoye MO, Adesiji YO. Bacteriological and Parasitoloficl assessment of vaginitis in pregnant women in Iseyin, Oyo State, Nigeria. Af J Clinical & Exp. Microbiology 2003;4(2): 116-126. https://doi.10.4314/ajcem.v4i2.7315

 Sallum EJ, Nouer DF, Klein MI, Goncalves RB, Machion L, Wilson Sallum A, Sallum EA. Clinical and microbiologic changes after removal of orthodontic appliances. Am J Orthod Dentofacial Orthop 2004;126(3):363–366.

https://doi.org/10.1016/j.ajodo.2004.04.017

- 36. You Y, Son G, Kwon EJ, Park MH, Lee K, Young Ju Kim YJ. New insight into the analysis of amniotic fluid microflora using 16S rRNA gene sequencing. JMM Case Reports 2016; DOI 10.1099/jmmcr.0.005021.<u>https://doi.org/10. 1099/jmmcr.0.005021</u>
- 37. Kook JK, Park SN, Lim YK, Cho E, Jo E, Roh H, Shin Y, Paek J, Kim HS, Kim H, Shin JH, Chang YH Genome-Based Reclassification of *Fusobacterium nucleatum* Subspecies at the Species Level. Curr Microbiol 2017;74(10):1137-1147. <u>https://doi.org/10.1007/s00284-017-1296-9</u>
- Sydsjö G, Lindell Pettersson M, Bladh M. et al. Evaluation of risk factors' importance on adverse pregnancy and neonatal outcomes in women aged 40 years or older. BMC Pregnancy Childbirth 19, 92 (2019). <u>https://doi.org/10.1186/s12884-019-2239-1</u>
- Shah PS; Knowledge Synthesis Group on Determinants of LBW/PT births. Parity and low birth weight and preterm birth: a systematic review and meta-analyses. Acta Obstet Gynecol Scand 2010 Jul;89(7):862-75.

https://doi.org/10.3109/00016349.2010.486 827

40. Koullali B, van Zijl MD, Kazemier BM, Oudijk MA, Mol BWJ, Pajkrt E, Ravelli ACJ. The association between parity and spontaneous preterm birth: a populationbased study. BMC Pregnancy Childbirth 2020;20(1):233.

https://doi.org/10.1186/s12884-020-02940w